---
layout: api
page_title: Identity - Secrets Engines - HTTP API
description: This is the API documentation for the Vault Identity secrets engine.
---

> [!IMPORTANT]  
> **Documentation Update:** Product documentation, which were located in this repository under `/website`, are now located in [`hashicorp/web-unified-docs`](https://github.com/hashicorp/web-unified-docs), colocated with all other product documentation. Contributions to this content should be done in the `web-unified-docs` repo, and not this one. Changes made to `/website` content in this repo will not be reflected on the developer.hashicorp.com website.

# Identity secrets engine (API)

This is the API documentation for the Vault Identity secrets engine. For general
information about the usage and operation of the Identity secrets engine, please
see the [Vault Identity documentation](/vault/docs/secrets/identity).

## API sections

- [Entity](/vault/api-docs/secret/identity/entity)
- [Entity Alias](/vault/api-docs/secret/identity/entity-alias)
- [Group](/vault/api-docs/secret/identity/group)
- [Group Alias](/vault/api-docs/secret/identity/group-alias)
- [Identity Tokens](/vault/api-docs/secret/identity/tokens)
- [Lookup](/vault/api-docs/secret/identity/lookup)
- [OIDC Provider](/vault/api-docs/secret/identity/oidc-provider)
- [MFA](/vault/api-docs/secret/identity/mfa)
